Author's response to reviews

Title: Workplace violence against physician and nurses in Palestinians public hospitals: A retrospective cross-sectional study

Authors:

Mohamad Kitaneh (ketaneh36@hotmail.com)
Motasem Hamdan (mhamdan@med.alquds.edu)

Version: 2 Date: 13 August 2012

Author's response to reviews: see over
Date: August 13, 2012

Subject: Re-submission of new research manuscript for publication entitled “Workplace violence against physician and nurses in Palestinians public hospitals: A retrospective cross-sectional study”

Dear Dr. Natalie Pafitis,
Executive Editor
BioMed Central

Enclosed is the reviewed research manuscript entitled “Workplace violence against physician and nurses in Palestinians public hospitals: A retrospective cross-sectional study” to be considered for publication in the BMC Health Services Research journal.

We have reviewed the manuscript based on the comments and recommendations of the section editor. In specific we have improved the discussion section. We added several new paragraphs to the section, whereby provided more interpretations and explanations of our result and linked them to available regional and international evidence. While doing that we have used three additional references [21-23].

We also added one sentence containing the result about the place where the violence occurred, in the end of the first paragraph- section entitled “Magnitude and consequence of violent events”. We can provide you with the original manuscript with track changes if needed.

We will appreciate if you could consider the manuscript for the peer review process. We will be at your disposal for any further information or amendments that you may require.

On behalf of the authors,
Sincerely yours,

Motasem Hamdan, PhD.

Dean, Faculty of Public Health,
Al-Quds University
Jerusalem, P.O.Box 51000
Email: mhamdan@med.alquds.edu